Author's response to reviews

Title: Reported food intake and distribution of body fat: A repeated cross-sectional study

Authors:

Benno Krachler (benno.krachler@medicin.umu.se)
Mats Eliasson (Mats.Eliasson@nll.se)
Hans Stenlund (hans.stenlund@epiph.umu.se)
Ingegerd Johansson (Ingegerd.Johansson@odont.umu.se)
Goran Hallmans (goran.hallmans@nutries.umu.se)
Bernt Lindahl (bernt.lindahl@medicin.umu.se)

Version: 4 Date: 12 December 2006

Author's response to reviews: see over
12 December 2006

Dear Editor of the Nutrition Journal,

Many thanks for your kind reply. We corrected the mistakes in table 1 that you pointed out and resubmitted.

Cover letter and main manuscript have been replaced by new versions (v4) whereas figures and additional table are unchanged (v2). We hope that these changes addressed your concerns and are happy to make further corrections, if necessary.

Yours

Benno Krachler

BENNO KRACHLER
DEPARTMENT OF MEDICINE
KALIX HOSPITAL,
SKOLGATAN 1
S-952 82 KALIX
SWEDEN
FAX: (+)46 923 76564
benno.krachler@medicin.umu.se